\documentclass[a4paper,notitlepage]{article}
\usepackage[T1]{fontenc}
\usepackage[latin1]{inputenc}
\newcommand{\strongemph}[1]{\textbf{#1}}
\newcommand{\litfmt}[1]{\texttt{#1}}
\title{}
\author{}
\date{}
\begin{document}
\maketitle
This goes into the ingress (\pageref{This.goes.into.the.ingress})\\
\tableofcontents

This test is \copyright{} 1851 E. A. Poe.  --- I think.
Making \label{smooth}smooth transitions is \label{hard}hard work.  It's not worth
hardening the \label{hard1}hard; you should look into means of making
things go more smoothly.

\label{This.goes.into.the.ingress}This goes into the ingress

Thank you for listening\ldots{} the \label{percussions}percussions.

smooth (\pageref{smooth})\\
hard (\pageref{hard})\\
hard (\pageref{hard1})\\
percussions (\pageref{percussions})\\

\end{document}
